Author's response to reviews

Title: Safety in Home Care: A Research Protocol for Studying Medication Management

Authors:

Patricia B Marck (patricia.marck@ualberta.ca)
Ariella Lang (ariella.lang@sympatico.ca)
Marilyn Macdonald (marilyn.macdonald@dal.ca)
Melissa Griffin (melissa.c.griffin@gmail.com)
Anthony Easty (tony.easty@uhn.on.ca)
Serena Corsini-Munt (serena.cm@gmail.com)

Version: 4 Date: 20 May 2010

Author's response to reviews: see over
Dear Editor;

I have carefully re-checked this manuscript and accepted all of your editorial changes.
I have also addressed your queries on page 21 of the manuscript re: references by revising the paragraph references as follows (see blue highlight sections below):

“The rigor of the interpretations that are developed in this study is strengthened by input at varying stages from research team members with expertise in one or more of nursing, medicine, pharmacy, HF engineering, healthcare ethics, law, patient advocacy, healthcare administration, home care practice, and qualitative and visual methods. Attention to rigor will be reflected by openly accounting for the impact of bias (personal, discipline/profession-based, et al.) on study results, recording clear documentation about the research-gathering contexts and procedures, and creating a logically auditable path from data collection to conclusion phases [49, 51]. The research team will also optimize
the trustworthiness of the study findings by means of the following measures; credibility, transferability, dependability, and confirmability [51, 52].”

Thank you for catching this loss of some of our intended references in the multiple revisions to the manuscript before our original submission. These references have also been added back in to the Reference List as follows:


Please do not hesitate to contact me should you have any further requirements to get this manuscript to press.

Thank you,

Patricia Marck RN, PhD